Author's response to reviews

Title: Job satisfaction trends during nurses' early career

Authors:

Trevor Murrells (trevor.murrells@kcl.ac.uk)
Sarah Robinson (sarah.robinson@kcl.ac.uk)
Peter Griffiths (peter.griffiths@kcl.ac.uk)

Version: 8 Date: 29 May 2008

Author's response to reviews:

To the Editor-in-Chief

Dear Dr Norton

The proof has been read and a few minor alterations made.

Page 12.
Client care changed to Client Care

Page 21.
Other changed to later
(Relationships, Pay) now displayed in italics

Page 26.
6 years changed to seven years

Figure Legend
adjusted for moderating variables removed form Figure title.

No further proof reading is required.

Regards

Trevor Murrells
National Nursing Research Unit
King's College London
United Kingdom